Author's response to reviews

Title: Human Pregnane X receptor is expressed in human breast carcinomas. Potential heterodimer formation between hPXR and RXR-alpha

Authors:

  Isabel Conde (isabelcondemartin@yahoo.es)  
  Maria V T Lobo (mval.toledo@uah.es)  
  Javier Zamora (javier.zamora@hrc.es)  
  Julio Perez (juliop.marquez@uah.es)  
  Francisco J Gonzalez (fjgonzalez02@yahoo.es)  
  Emilio Alba (oncologia98@yahoo.com)  
  Benito Fraile (benito.fraile@uah.es)  
  Ricardo Paniagua (ricardo.paniagua@uah.es)  
  Maria I Arenas (misabel.arenas@uah.es)

Version: 5  Date: 28 May 2008

Author's response to reviews: see over
LETTER EXPLAINING THE FORMATING CHANGES MADE IN THE FINAL VERSION OF THE MANUSCRIPT

1. Author list.
The full stops from the initials have been removed.

2. Affiliations and email addresses – Please list these separately.
They have been listed separately and the country has been included in all of them.

3. Title. Please use sentence case in the title.
The title has been modified conveniently.

4. Keywords – Please remove these from the abstract.
The keywords have been deleted.

5. Header – Please remove ‘Conde et al. PXR in breast cancer’ from the top of each page.
The header has been eliminated.

6. Introduction – Please rename this section ‘Background’.
This section has been renamed.

7. Table orientation – we prefer tables and figures to fit onto a portrait A4 page.
All the tables have now orientated onto a portrait A4 page.

8. Figure legends – Please make sure that each figure has it own legend. Currently figures 4 and 5 have the same legend.
Figures 4 and 5 have now their own legend.

9. Figure cropping – It is important for the final layout of the manuscript that the figures are cropped as closely as possible.
All the figures are cropped adequately and saved at maximum quality in JPEG files.